Author's response to reviews

Title: A three arm randomised controlled trial comparing Gonadotrophin Releasing Hormone (GnRH) agonist long regimen versus GnRH agonist short regimen versus GnRH antagonist regimen in women with a history of poor ovarian response undergoing in vitro fertilisation (IVF) treatment: Poor responders intervention trial (PRINT)

Authors:

Sesh K Sunkara Dr (sksunkara@hotmail.com)
Arri Coomarasamy Dr (arricoomar@blueyonder.com)
Yakoub Khalaf Mr (yakoub.khalaf@kcl.ac.uk)
Peter Braude Professor (peter.braude@kcl.ac.uk)

Version: 4 Date: 17 December 2007

Author's response to reviews: see over
Assisted Conception Unit,
Thomas Guy House,
Guy’s Hospital,
Guy’s & St Thomas’ NHS Foundation Trust,
London, SE1 9RT, UK.

Editorial Production Team
Bio Med Central.

17th December 2007

Dear Editorial Team,

Re: MS 1346048711155159.

Title: A three-arm randomised controlled trial comparing Gonadotrophin Releasing Hormone (GnRH) agonist long regimen versus GnRH agonist short regimen versus GnRH antagonist regimen in women with a history of poor ovarian response undergoing in vitro fertilisation (IVF) treatment: Poor responders intervention trial (PRINT).

The following formatting changes have been incorporated in the new version of the manuscript as per your instructions.

Full-stop removed from the end of the title.

Title page: the title page has been structured as per instructions.

Abstract: the abstract in the manuscript has been matched exactly with what is in the submission process.

Section headings: Unnecessary capitals have been removed.

Website address: Website has been removed from the text, and cited in the main reference list.

Acknowledgements section: an acknowledgement section has been added at the end of the manuscript text, directly preceding the reference list. The source of funding for the study has been listed.

Figures in text: figures have been excluded from the manuscript file. The figures have been uploaded without the title as instructed.
Additional files: a section titled ‘additional files’ at the end of the manuscript has been formatted as instructed. All additional files have been mentioned in the text and referred as instructed.

Thank you
Yours sincerely

Dr SK Sunkara MRCOG
(On behalf of the authors)